Title: Differential survival following trastuzumab treatment based on quantitative HER2 expression and HER2 homodimers in a clinic-based cohort of patients with metastatic breast cancer.

Authors:

Masakazu Toi (toi@kuhp.kyoto-u.ac.jp)
Jeff Sperinde (jspirinde@monogrambio.com)
Weidong Huang (whuang2@monogrambio.com)
Shigehira Saji (ss-saji@wa2.so-net.ne.jp)
John Winslow (jwinslow@monogrambio.com)
Xueguang Jin (xjin@monogrambio.com)
Yuping Tan (ytan@monogrambio.com)
Shinji Ohno (sohno@nk-cc.go.jp)
Seigo Nakamura (seigonak@luke.or.jp)
Hiroji Iwata (hiwata@aichi-cc.jp)
Norikazu Masuda (nmasuda@alpha.ocn.ne.jp)
Kenjiro Aogi (kaogi@shikoku-cc.go.jp)
Satoshi Morita (smorita@urahp.yokohama-cu.ac.jp)
Christos Petropoulos (cpetropoulos@monogrambio.com)
Michael Bates (mbates@monogrambio.com)

Version: 4 Date: 24 December 2009

Author's response to reviews:

We added the sentence below:

This study protocol was approved by the ethical committee of each institution and informed consent from the patients was taken.